Author’s response to reviews

Title: Clinical implications of granulomatous inflammation detected by endobronchial ultrasound transbronchial needle aspiration in patients with suspected cancer recurrence in the mediastinum

Authors:

Marcus P Kennedy (mpkennedy@uams.edu)
Carlos A Jimenez (cajimenez@mdanderson.org)
Ashwini D Mhatre (admhatre@mdanderson.org)
Rodolfo C Morice (rmorice@mdanderson.org)
Georgie A Eapen (geapen@mdanderson.org)

Version: 3 Date: 15 February 2008

Author’s response to reviews:

DEAR EDITOR, WE HAVE MADE ADJUSTMENTS TO THE MANUSCRIPT AS LISTED BELOW IN BLOCK CAPITALS.

Formatting changes requested
-------------------------------------------
Please remove the header from the manuscript ¿MS 1310684991158313 Journal of Cardiothoracic Surgery¿
REMOVED
Please remove the following text from the title page: ¿Title Page¿
REMOVED
Title ¿ Please remove the full stop or period from the end of the manuscript title. Please ensure the manuscript title is identical in both the manuscript and submission system.
REMOVED, BOTH TITLES ARE IDENTICAL
Qualifications - Please remove the authors’ qualifications/titles (e.g. PhD / MD / DR / Head of xx) from the manuscript and submissions system.
REMOVED
Short title - Please remove the short title from the title page.
REMOVED
Word count - Please remove all word counts from the manuscript.
REMOVED
Author email addresses ¿ If possible please give a different email address for each author, on the title page, beneath the corresponding author details.
Please link authors to their email addresses using their initials, for example:
ABC: Myemail@address.com
INCLUDED

Materials and Methods. Please rename this section to `Methods`.
REMOVED

Competing interests. Please include a section entitled Competing interests. Please write ‘The authors declare that they have no competing interests’ in this section.
INCLUDED

Authors’ contributions - Would you like to indicate the authors’ contributions for the manuscript? If so please place a section called Authors’ contributions between Competing interests and Acknowledgements.

For the Authors’ contributions we suggest the following kind of format before the Acknowledgements (please use initials to refer to each author’s contribution): For example, AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

NOT REQUIRED

Acknowledgements - Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study. Place the section directly before the references in the main body of the text.

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

NOT NEEDED

Table title and legend - The table title should appear above the table. The legend (or key) should appear below the table.
CORRECTED

Tables. Please reposition the tables below the figure legends.
CORRECTED